Author's response to reviews

Title: Infratentorial Benign Cystic Meningioma Mimicking A Hemangioblastoma Radiologically & A Pilocytic Astrocytoma Introperatively: Case Report

Authors:

Tan Kheng Guan (devaraj77@gmail.com)
Devaraj Pancharatnam (neri_mala@yahoo.com)
Hari Chandran Thambinayagam (chandran.hari18@gmail.com)
Teoh K.H (devaraj77@gmail.com)
Gnana Kumar P Gnanasuntharam (gnanak_rad@um.edu.my)
Dharmendra Ganesan (drdharmen@gmail.com)

Version: 4 Date: 9 October 2012

Author's response to reviews: see over
09/10/2012

To,

The Journal of Medical Case Reports Editorial Team

Title: Infratentorial Benign Cystic Meningioma Mimicking A Hemangioblastoma Radiologically & A Pilocytic Astrocytoma Introperatively: Case Report

Firstly I would like to thank you for considering our manuscript. We have analysed the comments by your reviewers. Listed below is the reply to the Editorial’s Comment.

Comment from the Editorial Office

Please refer to the patient as “30 year old Chinese female” in both the abstract and the main case presentation section

Reply

We have made the relevant changes as per requested in the case presentation section.

A 30 years old Chinese female lady, with a childhood history of acute lymphoblastic leukemia, currently in remission, presented to the Neurology clinic with a 3 weeks history of worsening headaches, giddiness and vomiting. The neurological examination revealed positive cerebellar signs in keeping with an infratentorial space occupying lesion. (Line 77-80)

We sincerely hope that these changes will be accepted and this manuscript accepted.
Thank you,

Yours sincerely,

…………………………………
Dr. Devaraj Pancharatnam
MBBS(Malaya), MRCSEd, MSurg (Malaya)
Senior Lecturer,
Division of Neurosurgery, Department of Surgery,
Faculty of Medicine, University Malaya,
Kuala Lumpur, Malaysia.